Author's response to reviews

Title:A Delphi survey to determine how educational interventions for evidence-based practice should be reported. Stage 2 of the development of a reporting guideline.

Authors:

Anna C Phillips (anna.phillips@mymail.unisa.edu.au)
Lucy K Lewis (lucy.lewis@unisa.edu.au)
Maureen P McEvoy (maureen.mcevoy@unisa.edu.au)
James Galipeau (jgalipeau@ohri.ca)
Paul Glasziou (Paul_Glasziou@bond.edu.au)
Marilyn Hammick (m.hammick@gmail.com)
David Moher (dmoher@ohri.ca)
Julie K Tilson (tilson@usc.edu)
Marie T Williams (marie.williams@unisa.edu.au)

Version:2Date:14 February 2014

Author's response to reviews:

Minor spelling errors were noted in the original submission of the manuscript which have now been corrected.